Author's response to reviews

Title: Human pericardium graft in the management of bleb's complication performed in childhood: a case report

Authors:

Dimitris Papaconstantinou (dpapaconstantinou@hotmail.com)
Ilias Georgalas (igeorgalas@yahoo.com)
Sergios Taliantzis (sergiotali@hotmail.com)
Chrysanthi Koutsandrea (ancode@otenet.gr)
Ioannis Ladas (idladas@gmail.com)
Gerassimos Georgopoulos (glaucomanet@aim.com)

Version: 6 Date: 6 September 2011

Author's response to reviews: see over
Cover letter

Dear editor

Reviewer Kazuyuki Hirooka

In line 1 of the abstract to… has been changed to To… and city and country of New World Medical has been added in page 4 line 17.

All editorial requests have been corrected.

The name of the hospital that granted ethical approval has been included and this statement has been moved from the competing interests to the case presentation section.

A separate Consent section has been included

Reformation of abstract’s section has been made

Changes in the manuscript are coloured and underlined

Sincerely yours

Taliantzis Sergios